

Schistosoma haematobium and Schistosoma bovis first generation hybrids undergo gene expressions changes consistent with species compatibility and heterosis

Eglantine Mathieu-Bégné, Julien Kincaid-Smith, Cristian Chaparro, Jean-François Allienne, Olivier REY, Jérôme Boissier, Eve Toulza

▶ To cite this version:

Eglantine Mathieu-Bégné, Julien Kincaid-Smith, Cristian Chaparro, Jean-François Allienne, Olivier REY, et al.. Schistosoma haematobium and Schistosoma bovis first generation hybrids undergo gene expressions changes consistent with species compatibility and heterosis. PLoS Neglected Tropical Diseases, 2024, 18 (7), pp.e0012267. 10.1371/journal.pntd.0012267 hal-04639136

HAL Id: hal-04639136 https://hal.inrae.fr/hal-04639136v1

Submitted on 2 Aug 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial 4.0 International License



Citation: Mathieu-Bégné E, Kincaid-Smith J, Chaparro C, Allienne J-F, Rey O, Boissier J, et al. (2024) *Schistosoma haematobium* and *Schistosoma bovis* first generation hybrids undergo gene expressions changes consistent with species compatibility and heterosis. PLoS Negl Trop Dis 18(7): e0012267. https://doi.org/10.1371/ journal.pntd.0012267

Editor: James Cotton, University of Glasgow, UNITED KINGDOM

Received: May 22, 2023

Accepted: June 3, 2024

Published: July 2, 2024

Copyright: © 2024 Mathieu-Bégné et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Raw sequences used in this study have been submitted to the Sequence Read Archive under the BioProject PRJNA491632. Further data and scripts necessary to replicate the study are available at https://doi.org/10.6084/m9. figshare.22109159.

Funding: This work has been funded by the French Research National Agency (project HySWARM, grant no. ANR-18-CE35-0001, to EMB). The

RESEARCH ARTICLE

Schistosoma haematobium and *Schistosoma bovis* first generation hybrids undergo gene expressions changes consistent with species compatibility and heterosis

Eglantine Mathieu-Bégné ^{1‡}, Julien Kincaid-Smith^{2,3‡}, Cristian Chaparro², Jean-François Allienne², Olivier Rey², Jérôme Boissier², Eve Toulza²*

1 Department of Environmental Sciences, Zoology, University of Basel, Basel, Switzerland, 2 IHPE, Université de Montpellier, CNRS, Ifremer, Université Perpignan Via Domitia, Perpignan, France, 3 CBGP, IRD, CIRAD, INRAE, Institut Agro, Université de Montpellier, Montpellier, France

† These authors share first authorship on this work. * eve.toulza@univ-perp.fr

Abstract

When two species hybridize, the two parental genomes are brought together and some alleles might interact for the first time. To date, the extent of the transcriptomic changes in first hybrid generations, along with their functional outcome constitute an important knowledge gap, especially in parasite species. Here we explored the molecular and functional outcomes of hybridization in first-generation hybrids between the blood fluke parasites Schistosoma haematobium and S. bovis. Through a transcriptomic approach, we measured gene expression in both parental species and hybrids. We described and quantified expression profiles encountered in hybrids along with the main biological processes impacted. Up to 7,100 genes fell into a particular hybrid expression profile (intermediate between the parental expression levels, over-expressed, under-expressed, or expressed like one of the parental lines). Most of these genes were different depending on the direction of the parental cross (S. bovis mother and S. haematobium father or the reverse) and depending on the sex. For a given sex and cross direction, the vast majority of genes were hence unassigned to a hybrid expression profile: either they were differentially expressed genes but not typical of any hybrid expression profiles or they were not differentially expressed neither between hybrids and parental lines nor between parental lines. The most prevalent profile of gene expression in hybrids was the intermediate one (24% of investigated genes). These results suggest that transcriptomic compatibility between S. haematobium and S. bovis remains quite high. We also found support for an over-dominance model (over- and under-expressed genes in hybrids compared to parental lines) potentially associated with heterosis. In females in particular, processes such as reproductive processes, metabolism and cell interactions as well as signaling pathways were indeed affected. Our study hence provides new insight on the biology of Schistosoma hybrids with evidences supporting compatibility and heterosis.

funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

When two species, composed of different parental genetic material, manage to produce a viable offspring, new allelic interactions might arise. We do not know much about how genes are expressed in such hybrids compared to their parental species, especially for parasite species. Here we investigated gene expressions in first generation hybrids of two blood fluke parasites: Schistosoma haematobium and S. bovis. We quantified and assigned in five typical profiles genes expressed in first generation hybrids (i.e., intermediate between the parental expression levels, over-expressed, under-expressed, or expressed like one of the parental lines) and investigated their associated biological processes. Gene expressions in hybrids revealed a substantial fraction of genes being unassigned (once the sex and the direction of the cross from which a hybrid resulted were controlled) which were differentially expressed genes but not typical of any expression profiles or were not differentially expressed neither between hybrids and parental lines nor between parental lines. Among assigned genes, we recorded a predominance of genes being expressed at an intermediate level in hybrids compared to both parental species, hence highlighting the compatibility between the two species. Interestingly, we found, in particular in female hybrids, genes over- and under-expressed compared to parental lines. Those synergetic expression profiles are usually associated with heterosis in a model called over-dominance. At the functional level, our results provide evidence for compatibility between parental lines and potential heterosis in hybrids with biological processes expressed in hybrids involved in signaling pathways, energy intake, and reproduction.

Introduction

Species definition has for long fueled debates among biologists due to the difficulty to draw clear boundaries between such biological entities [1]. The study of hybridization offers the possibility to shed light on the process of species formation. So far, hybridization has been mostly studied in terms of populations structure, genome structure and phenotypic outcomes [2,3]. However, the molecular mechanisms underlying the viability and performance of hybrids still constitute an important knowledge gap, especially regarding empirical evidence. In particular, there are significant research opportunities in studying the gene expression, epistasis and epigenetics marks of hybrids that arise from non-model parental lines (i.e., ecologically realistic hybrids) that have to date been less studied than model laboratory species or species of economic interest [4–6].

The study of hybridization has been mostly motivated by two extreme phenotypes in hybrids, namely hybrid incompatibilities and heterosis (or hybrid vigor, [7]). Hybrid incompatibilities describes the case when the hybrid phenotype cannot be maintained or achieved due to conflicts in the expression of each parental species genomes (i.e., *genome clash*, [7]). At the other end of the spectrum, heterosis is a phenotype characterized by higher performance of the hybrids compared to parental lines in term of growth, size or reproductive potential [8,9]. Heterosis may provide a particular benefit in farmed species because it increases their yield, explaining why a substantial part of the literature on heterosis has focused on species of agroe-conomic interest. Most first-generation hybrids are sterile or non-viable at all due to parental genome conflicts, though when viable they usually display vigor on at least some traits [7]. Interestingly, if heterosis can emerge in the first generation of hybridization, this phenotype is not maintained through subsequent generations that usually show a decrease in heterozygosity

(which get them closer to the parental phenotype) and may ultimately display hybrid breakdown (i.e., weaker performance than both parental lines, [10]). Hence the "hybrid phenotype" is a dynamic process throughout hybrid generations that depends on the interaction of alleles that are brought together at each generation [8].

Similarly, the process of hybridization does not occur equally among species since permeability toward hybridization may drastically vary between genomic backgrounds. Even after millions of years some species still display significant levels of admixture whereas other species, although close genetically, have evolved strong reproductive isolation mechanisms that prohibit hybridization either at early stages of reproduction (pre-zygotic barriers) or along the hybrid lifespan (post-zygotic barriers, for instance when hybrids die prematurely or are sterile) [11]. Once successful fertilization has occurred, producing a viable hybrid zygote depends on parental genomes' interaction and environmental selection. For instance, differences in the karyotype generally results in non-compatible genome interaction and non-viable offspring [12]. At the gene level, when two species hybridize, some alleles that have not been brought together might interact to result in an hybrid phenotype [13]. Depending on the degree of genetic divergence between parental lines (and how conserved their transcription factors are), hybrids might form and prosper or might just not be viable [13]. Some theoretical gene-level models have been developed to explain exacerbated hybrid phenotypes such as heterosis (eg., dominance and over-dominance models which expect either selective expression of the most advantageous parental allele or synergetic expression of parental alleles respectively [9]). Interestingly, incompatibility is also expected to be associated with more over- and under-expressed genes in hybrid than intermediate level of expression compared to parental lines, [13]. However, those theoretical expectations remain mostly explored in species of agronomic interests and provided sometime contradictory results [13]. For species hybridizing naturally with significant introgression levels, even description of gene expression patterns remains to date rarely documented.

Among naturally hybridizing species, there is a growing interest in parasites and/or their vectors since their geographic range can be affected by global changes (including human migration, habitat loss, global trade and climate change), hence bringing together pathogens that were previously not in contact [14]. This is particularly worrying due to the potential for exacerbated virulence in hybrid parasites [15]. Viable hybridization in parasites often results in stronger detrimental effects on their hosts, but also broader vector/host range, which carry the potential for zoonotic transmission when parasites of wild or farmed animals interact with human parasites [14]. For instance, introgression between two subspecies of Trypanosoma brucei have been shown to be associated with different levels of virulence [16]. Similarly, the high competence of malaria vectors, as revealed by their broad definitive host spectrum that encompassed various mammals and birds species, is thought to result from genetic admixture between Anopheles species which is inferred from their high levels of introgression [14,17,18]. To date most studies have focused on identifying and quantifying parasite hybrids in the wild. Although necessary to assess the risks associated with parasite hybridization, it is currently very rare for studies to focus on the molecular bases of the hybrid phenotypes in parasites (but see [19]).

Schistosoma haematobium and S. bovis are two blood fluke parasites for which hybrid vigor has been documented in the laboratory [20] and is suspected to have impacts on human morbidity in the field [21]. Schistosomes are the etiologic agents of Schistosomiasis (or Bilharziasis) a Neglected Tropical Disease affecting over 240 million people in the world [22]. These gonochoric parasites have a two-host life cycle including a mollusc intermediate host in which asexual multiplication occurs, and a mammalian definitive host in which sexual reproduction (separate sexes) takes place. In the latter host, parasite species co-infection can lead to

interbreeding and the production of viable hybrid progeny [23,24]. Actually, some recent studies have revealed various levels of introgression notably in S. haematobium from different other species of Schistosoma including S. bovis and most likely following a scenario of ancient introgression [25–27]. However, the question of potential ongoing hybridization is still open and more knowledge is required on those potential hybrids to better understand their impact on future transmission dynamics of schistosomiasis. Several field or experimental viable interspecific crosses have indeed been evidenced among the Schistosoma genus [28]. Depending on the phylogenetic distance between the interacting species, the genetic background of the resulting progeny varies between parthenogenic (hence non-viable introgressed individuals) to fully viable introgressed individuals [29]. Schistosoma haematobium and S. bovis are sister species amongst the haematobium clade [30]. Schistosoma haematobium is specialized toward humans as definitive hosts, while S. bovis is a livestock and rodent parasite. They also display differences in their tropism since S. haematobium is urogenital while S. bovis is located in the mesenteric vein system. However, despite apparent host-specific barriers, hybrids between S. haematobium and S. bovis are observed in the field with common occurrences in several African countries including Senegal, Benin, Mali, Niger, Cameroon, Ivory Coast or Nigeria [27,31–37]. Schistosoma haematobium x S. bovis hybrids were also involved in the European outbreak that occurred in Corsica in 2013 and where infections are still persisting and expanding to nearby rivers [38,39]. Both for human health impacts and epidemiological aspects, schistosome hybrids raise worrying issues. Schistosome first-generations hybrids tend to produce both greater number of eggs and sometimes larger or longer eggs than parental forms [27,28,40,41] and the presence of eggs trapped in human tissues are responsible for the physiopathology of the disease. Hybridization among schistosomes is hence expected to affect parasite-induced pathology and transmission with important consequences for public health and disease control. Past studies on schistosome hybrids have focused on describing their distribution and occurrence in natural populations, here, we attempt to provide new insight on some of the molecular changes occurring following first-generation hybridization between S. haematobium and S. bovis.

In this study we aimed at characterizing gene expression levels and molecular pathways that vary in S. haematobium and S. bovis first generation hybrids compared to parental species. Relying on a genome-guided transcriptomic approach we first assembled transcripts from S. haematobium, S. bovis and their hybrids and compared their gene expression levels using S. haematobium genome as reference. We categorized hybrid expression profiles by comparing them to both parental lines. Hence, we distinguished five expression profiles of the hybrids namely: over-expressed, under-expressed, intermediate (i.e., intermediate between the parental expression levels also referred as additive expression, [42,43]), S. haematobium-like and S. bovis-like expression profiles. Finally, among genes belonging to each category of expression profiles we tested whether some biological processes were significantly enriched (i.e. over-represented). We expect to find evidence for conserved compatibility between S. haematobium and S. bovis due to their introgression history that could translate into a high portion of genes with intermediate levels of expression between parental lines (i.e., no synergetic or dominance effects). However, based on current evidence on heterosis in cultivated species and model organisms, we could also expect to observe some genes having a non-intermediate profile (over- or under-expressed and parent-like, known as over-dominance and dominance model of gene expression respectively) that may sustain an increased expression of hybrid vigour traits (such as reproduction or growth) when comparing the hybrids to at least one of the parental species. Note that here we refer to heterosis as the enhancement of some traits resulting of the mixing of genetic contribution from each parent in their offspring, whether the parental lines consist of two different species or two entities distinct enough to induce a

substantial outbreeding event in their first-generation offspring. We hence consider that heterosis can apply for trait enhancement in *S. haematobium x S. bovis* offspring even in presence of introgression between parental lines as long as the two are distant enough and the introgression events ancient.

Material and methods

Ethics statement

Housing, feeding and animal care, including experiments on animals were carried out according to the national ethical standards established in the writ of 1 February 2013 (NOR: AGRG1238753A) setting the conditions for approval, planning and operation of establishments, breeders and suppliers of animals used for scientific purposes and controls. The experiments carried out for this study were approved and provided a permit A66040 for animal experimentation by the French Ministry of Agriculture and Fishery (Ministere de l'Agriculture et de la Peche), and the French Ministry for Higher Education, Research and Technology (Ministere de l'Education Nationale de la Recherché et de la Technologie). The investigator has the official certificate for animal experimentation, obtained from both ministries (Decret n ° 87/848 du 19 octobre 1987; authorization number 007083). Parasite collection campaign is described in [37]. Specifically, the campaign was authorized by the National Ethics Committee of Cameroon (No. 2019/01/UDM/PR/CIE), the Ministry of Public Health and the Ministry of Basic Education and within each surveyed schools, by local administrative authorities, school inspectors, directors and teachers [37]. All patients found to be positive were treated in accordance with WHO recommendations.

Study model

Parasite parental species and reciprocal experimental crosses. Schistosoma haematobium was initially recovered from infected patients in the Southeast part of Cameroon (Barombi Kotto lake, 4°28'04"N. 9°15'02"W) in 2015 [37]. Urine was collected during a Bilharziasis screening campaign and oral consent was obtained for each patient [37]. Eggs from positive urine samples were collected and we exposed local intermediate hosts (Bulinus truncatus) to five miracidia before they were transferred to the laboratory. Parasites were further maintained through completion of their cycle using Golden hamsters (Mesocricetus auratus) as definitive hosts and *B. truncatus* as intermediate host [24]. Because *S. haematobium* has recently been shown to be introgressed with S. bovis at various degrees [25,26] we took several precautions to ensure that the S. haematobium strain we used was as pure as possible (i.e., if introgression occurred it was far enough in the past to consider that the offspring between S. haematobium and S. bovis would undergo a significant outbreeding event at the first generation). First, this strain of S. haematobium was specifically collected in Barombi Kotto because of the absence of livestock hence limiting interspecific interactions with other Schistosoma species (in particular with S. bovis, [44,45]). To further test for the purity of S. haematobium, 3 cercariae from 13 infected molluscs were genotyped using 11 microsatellite markers and were compared to S. haematobium miracidia from human [37] and adult S. bovis collected from cattle [45] in Cameroun. All S. haematobium individuals clustered apart from the S. bovis samples with no indication of potential hybrids (S1 Fig). We further sequenced the COI gene from 35 schistosomes of this Barombi S. haematobium strain and all those sequences were referenced as S. haematobium sequences only (no sequence similar to other schistosome species were found, COI sequences are available in GenBank under accession numbers PP537975-PP538009). On the other hand, S. bovis was recovered from the Spanish laboratory of parasitology of the Institute of Natural Resources and Agrobiology in Salamanca, and

maintained in the laboratory with both *Bulinus truncatus* and *Planorbarius metidjensis* sympatric intermediate hosts and on *M. auratus* definitive hosts [46].

Starting with experimental crosses, we first created a full-sibling inbred line of *S. haemato-bium* because of its potential high genetic diversity compared to the *S. bovis* isolate maintained in laboratory for several decades (i.e., inbred isolate maintained in the lab since the 71's). A high genetic diversity in one of the parental lines would indeed impede our ability to detect differentially expressed genes due to a higher variance between samples. We hence crossed a single male with a single female after molecular sexing of the parasite larvae [47]. Briefly, batches of *B. truncatus* were individually infected with a single miracidium of *S. haematobium*, allowing the development of clonal male or female genotype. At patency (i.e., 55 days after exposure) cercariae genotypes emitted from individual molluscs were sexed and parasites from two molluscs (one infected with a male genotype and one with a female genotype) were used to infect hamsters (300 cercariae of each sex) by surface application method for one hour. The resulting offspring constituted our full-sibling inbred line used for subsequent experiments. Methods employed for mollusc and rodent infections and parasite collection were described previously [48–50].

To recover parental species adult worms and assess their gene expression profiles, we infected six hamsters with homospecific combinations of 600 cercariae (300 cercariae of each sex) of *S. haematobium*, or *S. bovis*. Briefly, for both parental species, a pool of *miracidia* obtained from homospecific crosses was used to infect molluscs (5 *miracidia* per mollusc). Fifty-five days after exposing the molluscs to the parasites, pools of 600 *cercariae* from infected molluscs were used to infect hamsters. Three months after infection hamsters were euthanatized and adult worms were recovered by hepatic perfusion.

Laboratory *S. haematobium x S. bovis* hybrids were produced by exposing molluscs (*B. truncatus* or *P. metidjensis*, respectively) to a single miracidium of either parasite species and using the resulting single sex clonal pools of *cercariae* of each species to infect hamsters with mixed combinations of parasites in equal proportions (Fig 1). *Schistosoma haematobium* and *S. bovis* were reciprocally crossed to produce F1 (female *S. haematobium* x male *S. bovis*) and F1'(female *S. bovis* x male *S. haematobium*) hybrids. Batches of six hamsters per combinations were infected using 300 female *S. haematobium* and 300 male *S. bovis cercariae* (F1 cross; n = 6) or 300 male *S. haematobium* and 300 female *S. bovis cercariae* (F1' cross; n = 6). Three months and twenty days after infection, hamsters containing the adult parents (heterospecific couples) and hybrid egg progenies were euthanized.

In order to recover first generation reciprocal F1 and F1' hybrid adult worms and evaluate their expression profiles, hybrid eggs were hatched and hybrid *miracidia* were used to infect *B. truncatus* snails (n = 5 *miracidia* per mollusc). After the prepatent phase (~35 days) the F1 and F1' cercariae were used to infect hamsters (n = 7 hamsters for each cross) with pools of 600 cercariae of unknown sex, and F1 and F1' adult worms were collected from portal perfusion at 90 days after infection. A graphical representation of the experimental cross design is shown in Fig 1.

RNA extraction and transcriptome sequencing procedure

Using a binocular microscope and a small paintbrush we separated the paired adult worms according to their sex to avoid any contamination from the other sex. We constituted pools of 10–12 individuals per condition (*S. haematobium*, *S. bovis*, F1 and F1') and sex in three biological replicates representing a total of 24 samples (4 conditions x 2 sexes x 3 replicates). These samples were directly flash frozen in liquid nitrogen, and stored at -80°C. Total RNA extractions were performed using the TRIzol Thermo Fisher Scientific protocol (ref: 15596018)

90 days

F1' clonal cercariae

Recovering F1' hybrid

adults and F2' eggs



Cross 1 = F1 (♀ S. haematobium x ♂ S. bovis)



eggs and

35 da

https://doi.org/10.1371/journal.pntd.0012267.g001

Sexing procedure

of cercariae

S. bovis

Monomiracidial

infection of

slightly modified as each reagents volume was halved. Briefly, pools of frozen adult worms in 2ml microtubes were ground with two steel balls using Retsch MM400 cryobrush (2 pulses at 300Hz for 15s). After extraction following manufacturer's protocol, total RNA was eluted in 44 μ l of ultrapure water. DNase treatment was then performed using Thermofisher Scientific Turbo DNA-free kit (ref: AM1907). RNA was then purified using the Qiagen RNeasy mini kit (ref: 74104) using manufacturer's protocol and eluted in 42 μ l of ultrapure water. Quality and concentration were assessed by spectrophotometry with the Agilent 2100 Bioanalyzer system and using the Agilent RNA 6000 nano kit (ref 5067–1511). Library construction and sequencing were performed at the Génome Quebec platform. The TruSeq stranded mRNA library construction kit (Illumina Inc. USA) was used following the manufacturer's protocol with 300 ng of total RNA per sample. Sequencing data are available at the NCBI-SRA under the Bio-Project accession number PRJNA491632.

Recovering F1' · Hamsters infection with pools of

Transcriptome assembly

110 days

Hamsters infection with

parental lines

We conducted a reference-based transcriptome assembly using *S. haematobium* reference genome [51] and the RNA-sequencing data generated in this study. Note that although *S. haematobium* was used as a reference, recent genomic characterization of this strain suggests that in fact it contains introgressed *S. bovis* alleles in genomic regions spanning up to 100 kb

[52,53]. First reads were quality trimmed and adapter were removed using Trimmomatic (version 0.39 [54]) with default paired-end settings [54]. Reads were then mapped on the genome of *S. haematobium* using the splice aware mapper Hisat2 (version 10.3.0 [55]). Sam output were converted into bam, sorted and indexed using Samtools (version 1.16.1 [56]). In order to gather already referenced transcripts but also potentially new ones the software Stringtie (version 2.1.7 [57]) was used to assemble transcripts per sample while recovering intron/exon structure. Finally, Stringtie's output files from each sample were merged in ordered to obtain congruent transcripts IDs across individual assemblies using the option—merge from Stringtie. This last step resulted in a transcriptome assembly of 66,012 transcripts and 16,840 genes while the reference genome of *S. haematobium* contains 14,698 transcripts and 9,431 protein-coding genes. Our assembly was compared to the reference gene models of *S. haematobium* to ensure its quality using Gffcompare (version 0.11.2 [58], S1 and S2 Tables).

All bioinformatics calculations were performed at sciCORE (http://scicore.unibas.ch/) scientific computing center at University of Basel.

Quantification of gene expression and differential expression analysis

Gene expressions were quantified using the software FeatureCounts (version 2.0.3, from the Subread suite [59]) with mapped reads of each sample and the transcriptome assembly achieved in this study as input. A matrix of gene raw counts was outputted and used for further differential expression analysis conducted in R (version 4.0.0 [60]) and using the package DESeq2 (version 1.28.1 [61]). A brief inspection of our gene expression dataset was performed using a Principal Component Analysis and revealed that one F1 replicate suspiciously clustered apart from others and with the wrong sex cluster. We thus excluded this F1 replicate from further analysis (S2 Fig). We further filtered out genes with less than 10 counts across all samples hence resulting in a set of 15,770 analyzed genes over 35 samples (see results).

The R package DESeq2 that internally corrects the library size and allows calculating differential gene expression was used to determine gene expression profiles. Based on expression profiles we defined five categories of hybrid genes that were captured setting proper contrasts in DESeq2. The *under-expressed profile* refers to genes that were under-expressed in hybrids compared to both S. haematobium and S. bovis with a significance level defined at a False Discovery Rate of 5% (FDR<0.05). The over-expressed profile describes genes that are significantly over-expressed in hybrids compared to both *S. haematobium* and *S. bovis* (FDR<0.05). The intermediate profile encompasses genes that have an expression level between those of the parents, (i.e. genes that are significantly over-expressed in hybrids compared to one of the parental species while being significantly under-expressed compared to the other at a FDR<0.05). The haematobium-like profile describes genes that are significantly differentially expressed between S. haematobium and S. bovis (FDR<0.05 and Log2Fold Change (LFC)>1) but that are similarly expressed between hybrids and S. haematobium (Log2Fold Change (LFC)<0.05 and FDR>80%). Finally, the *bovis-like profile* refers to genes that are significantly differentially expressed between S. haematobium and S. bovis but that are similarly expressed between hybrids and S. bovis. The remaining unassigned genes included some for which the levels of expression did not differ significantly neither between parental lines nor between any parental line and hybrid offspring and/or genes with a significant differential expression but that did not fulfill the set of criteria (FDR thresholds and/or LFC) to be classified amongst the different hybrid expression profiles. To capture cross-specific and sex-specific gene expression patterns, hybrids expression profiles were quantified within each sex and each cross (i.e., F1 or F1'). Hence, four types of comparisons were done, namely F1 female vs S. haematobium and S. bovis female, F1' female vs S. haematobium and S. bovis female, F1 male vs S. haematobium

and *S. bovis* male, and finally F1' male *vs S. haematobium* and *S. bovis* male. Each of these comparisons allowed quantifying genes belonging to the five expression profiles as defined and described above.

Transcriptome annotation and functional analysis

In order to annotate our newly assembled transcriptome, we used previous annotations from S. haematobium reference genome in combination with a de novo approach using different programs hosted either on the Galaxy instance of the IHPE laboratory or of the European server [62,63]. First, we extracted nucleotide sequences of each transcript we assembled previously using Gffread ([58], Galaxy Version 2.2.1.3+galaxy0). We then translated our transcripts sequences using TransDeCoder with default parameters ([64], Galaxy Version 5.5.0+galaxy2) in order to extract all Open Read Frames (ORF). Over the 66,012 assembled transcripts, 51,428 contained an ORF. Translated sequences were then annotated with Interproscan ([65], version 5.59–91.0+galaxy3) using default parameters. The annotation file in gff3 format was then converted to a tsv format using AGAT [66] in order to extract Gene Ontology (GO) terms, gene products terms and associate them to their corresponding gene. Note that a gene could be associated with several gene products if they coded for several transcripts and/or if they contained several ORFs. Finally, we also transferred to our assembly the annotations from the reference genome when genes were matching. Our annotation hence resulted in 10,256 annotated genes (with a product or a GO term) over the total of 16,840 genes assembled (61% overall annotation rate, 37% annotation rate based on GO terms only). The annotation rate obtained was a little lower than that of S. haematobium's reference genome (48% for GO terms annotations [51]), which is likely due to novel assembled genes that were absent from annotation databases.

Regarding the functional analysis, we conducted classic exact fisher enrichment tests using the corresponding parameters in the Rank Based Gene Ontology Analysis suit of scripts [67] (https://github.com/z0on/GO_MWU; with default parameters, but absValue = 0.001, and input file coded in binary with "1" indicating that the gene belongs to the gene set and "0" that it does not) and run on the IHPE Galaxy instance. Exact fisher enrichment tests compared within each GO term the number of genes from a defined gene set to the number of genes from the reference to test whether this GO contains significantly more genes from the gene set compare to genes from the reference. P-values were further adjusted for multiple comparisons and significant GO terms were selected at a 5% FDR threshold. To ease results interpretation, we further selected non-redundant significant GO terms falling into biological processes and displayed Tree Map graphical representations using Revigo with list size set to small [68]. The functional analyses were conducted on gene sets of each hybrid expression profile categories (under- and over-expressed, intermediate, *S. haematobium*-like and *S. bovis*-like) independently for each hybrid cross (F1 and F1') and each sex (female and male).

Results

Reference transcriptome assembly

A total of 16,840 genes were assembled in our new transcriptome. According to the Gffcompare report that compared our transcriptome assembly to *S. haematobium* gene models, only a few features (i.e., exon, intron, intron chain, transcript or genes) which were found in the reference annotation were not exactly matching those of our assembly (high Sensitivity value, S1 <u>Table</u>). On the contrary our assembly presented features that were not found in *S. haematobium* gene models (precision values were lower notably at the transcript level, S1 <u>Table</u>). No gene or transcript from the reference were missing from our assembly (S2 <u>Table</u>) and 14,668 transcripts of the 14,698 reference transcripts had a perfect match with one of our assembled transcript (S2 Table). A total of 5095 of gene in our assembly were totally novel (none of their transcripts matched with those of the reference, S2 Table). We hence conducted further analyses on the 16,840 genes we assembled since they contained features specific to our input RNA-seq data.

Gene expression profiles

The Principal Component Analysis conducted on parent and hybrid samples revealed that one F1 replicate clustered apart from others and in the wrong sex cluster hence suggesting that this sample was probably contaminated, we thus excluded it from further analysis (S2 Fig). Sex was the main segregating factor (first axis of a PCA conducted on all gene expression, S2 Fig), and, as expected, F1 samples were clustering between S. haematobium and S. bovis samples, (S2 Fig). We categorized five typical gene expression profiles being, over-expressed, underexpressed, intermediate, S. haematobium-like or S. bovis-like (see Figs 2 and S3-S5). After removing 1,070 lowly expressed genes (i.e., transcripts with less than 10 counts across all samples) from the initial 16,840 genes of the transcriptome, 15,770 genes were considered in the gene expression analysis. Overall, 7,100 genes (45%) were assigned to a hybrid expression profile (i.e., under-expressed, over-expressed, intermediate or like one or the other parental species) and 8, 670 (55%) genes were not assigned to a particular hybrid expression profile (those genes were either differentially expressed between the hybrid offspring and parental lines but without satisfying the criterions of each expression profile or were genes that were not differentially expressed at all, neither between hybrids and parental lines nor between S. haematobium and S. bovis, Sheet A in S1 File). The most abundant expression profile was the intermediate profile (i.e., 3,723 genes, 24% of investigated genes) whereas the less abundant profiles were the S. haematobium-like profile and the S. bovis-like profile with 592 (4%) and 319 (2%) genes respectively (Fig 3A, Sheet A in S1 File). Over- and under-expressed profiles included 1,501 (10%) and 1,912 (12%) genes respectively (Sheet A in S1 File). Interestingly for over- and under-expressed profiles, a substantial part of those genes was found in females and more frequently in F1 cross than in F1' cross (see Fig 3A). Also, very few genes falling into each of the other expression profiles were common between crosses (i.e., between F1 and F1') and between sexes (i.e., between males and females, see S6 Fig). We found common genes between crosses and sexes mostly for intermediate profiles (8.1% of the genes having an intermediate profile, see S6 Fig). Specifically, 9.1% of genes belonging to the intermediate profile were both found in F1 and F1' females and 18.2% were both found in F1 and F1'males. Some over-expressed genes were found both in F1 and F1' females (7.3% of all over-expressed genes) and some under-expressed genes were found both in F1 and F1' females (17.2% of all underexpressed genes). On average 81% of genes were unassigned to a particular gene expression profile within each cross and sex (Fig 3B).

Functional analysis

Gene Ontology (GO) terms were found enriched in genes belonging to eight sets of gene expression profiles compared to the overall pool of gene considered (16,840 total genes). Precisely, GO terms were significantly enriched with genes belonging to intermediate expression profiles (F1 and F1' females, F1 and F1' males), over-expressed profiles (both F1 and F1' females) and under-expressed profiles (both F1 and F1' females, see Sheets B and C in <u>S1 File</u> and <u>Fig 4</u>). Conversely, no GO terms were found enriched with gene sets corresponding to *S. haematobium* and *S. bovis* like expression profiles or with gene sets corresponding to male over- and under-expressed profiles.



Fig 2. Heatmaps illustrating the gene clustering within each hybrid expression profiles and presenting as example F1 females (F1) compared to *S. haematobium* (SH) and *S. bovis* (SB) females: under and over-expressed profiles (a), intermediate profiles (b), *S. haematobium*-like profile (c) and *S. bovis*-like profile (d). See <u>S3-S5</u> Figs for heatmaps of expression profiles of F1' females, F1 males and F1' males.

https://doi.org/10.1371/journal.pntd.0012267.g002

GO terms enriched in genes belonging to intermediate gene expression profiles were quite congruent between cross and sexes and including core terms such as *DNA integration* and *Carbohydrate metabolic process* (Fig 4). GO terms enriched in over-expressed genes were also congruent between F1 and F1'crosses both including the core term *non-coding RNA metabolic process*. Interestingly the GO term *Reproductive process* was found enriched in over-expressed genes in F1' females but not in F1 females. Finally, GO terms enriched in under-expressed genes were once again shared between F1 and F1' females, both including terms such as *ana-tomical structure development, cell-adhesion, cell-cell adhesion*, and *multicellular organismal process*.



Fig 3. a) Barplot showing for males and females of each hybrid cross, the number of genes falling into the over-expressed, under-expressed, intermediate, *S. haematobium* (*S.h.*) and *S. bovis* (*S.b.*) like expression profiles. b) Pie charts showing the ratio between genes assigned to a gene expression profile and unassigned genes within each cross and sex.

https://doi.org/10.1371/journal.pntd.0012267.g003

It is worth noting, that when directly examining the annotated gene product belonging to intermediate expression profiles, a high number of genes associated with retrotransposon activity were found (See Sheet B in <u>S1 File</u>). Specifically, in the entire transcriptome 773 genes were tagged with the term *reverse transcriptase* as product. Up to 278 (36%) of these were genes belonging to intermediate expression profile, while 84 (11%), 50 (6%), 20 (3%) and 17 (2%) were found in over-expressed genes, under-expressed genes, *S. bovis*-like expressed genes and *S. haematobium*-like expressed genes respectively. Hence the GO term *DNA integration* in intermediate expression profiles is most likely linked to retro-transposon activity.

Discussion

In this study, we aimed at describing first generation *S. haematobium x S. bovis* hybrids compared to their parents at the transcriptomic level. We were able to assemble up to 16,840 genes and assign their expression profiles into five categories namely: intermediate, over- and underexpressed, *S. haematobium*-like and *S. bovis*-like. The intermediate expression profile was the most common, but both under- and over-expressed profiles appeared relatively abundant in F1 females (*S. haematobium* mother and *S. bovis* father) and to a lesser extent in F1' females (*S.*

Intermediate F1 females Intermediate F1' females Intermediate F1 males											
DNA integration Integration process Intermediate F1' males				DNA metabolic integration process	DNA integration	organop meta pro	prganophosphate metabolic process pro		carl metab	carbohydrate letabolic process	
DNA integration	DNA integration DNA integration DNA integration DNA integration DNA integration DNA integration		ate carbohydrate lic metabolic ^{IS} process		DNA metabolic	monocarboxylic acid metabolic process		nucleobase- containing small molecule metabolic process	generation of precursor metabolites		
DNA metabolic oprocess	nucleobase- containing sma molecule metabolic process	all ADP metabolic process		carbobydrate	process	ADP metabolic process			and energy		
small molecule metabolic process				atabolic process	small molecule metabolic process		carbohy derivat metabolic j	carbohydrate derivative netabolic process		carbohydrate catabolic process	
Over-expressed F1 females Under-expressed F1 females									ales		
amide metabolic process	netabolic cess tRNA metabolic process		protein folding		regulation of signaling to the signaling		regulation of response to stimulus g cell surface receptor signaling	microt based p	ubule- rocess	anatomical structure development	
ncRNA metabolic cai process meta		boxylic acid bolic process		nall molecule abolic process			pathway negative regulation of signal transduction	cytosk organi	eleton zation	microtubule- based mouvement	
regulation of catabolic process		lational iation ocess	electron transport chain				ar cell adhesion	n cell- adhe	-cell sion	monoatomic cation transport	
Over-expressed F1' females						Jnder-	expressed	d F1' fem	nales		
ncRNA DNA metabolic metabolic process ncRNA metabolic process tregulatio		re n of	reproductive	inorganic cation transmembrane transport		cell ac	cell adhesion		anatomical structure development		
DNA biosynthetic process		activity		process	multicellular organismal process		G protein receptor pat	G protein-coupled receptor signaling pathway		cell-cell adhesion	

Fig 4. Tree maps of non-redundant biological processes enriched in genes having an intermediate profile, in genes having an overexpressed expression profile and in genes having an under-expressed expression profile. Relevant sex and cross are indicated above each tree map.

https://doi.org/10.1371/journal.pntd.0012267.g004

haematobium father and *S. bovis* mother). While genes falling into each expression profile were cross- and sex-specific, the corresponding biological processes were congruent between crosses and sexes. We specifically showed an over-representation of *DNA integration* function in intermediate profiles that could be reflecting an enhanced retro-transposon activity as expected in hybrids. In addition, biological processes associated with heterosis such as *Reproductive process* and more generally *Development* were represented among over- and under-expressed gene expression profiles. We hence provide new insights on the molecular machinery expressed within first generation hybrids while providing some explanation for the progeny viability and potential heterosis arising from *S. haematobium x S. bovis* hybridization.

Over the 15,770 genes tested we found that up to 7,100 genes were differentially expressed in Schistosoma hybrids compared to at least one of the parental species suggesting that some changes in gene expression patterns occurred in S. haematobium x S. bovis first generation hybrids. First generation hybrids benefit from an entire new set of alleles brought from the combination of the two parental species. This can lead to a "genomic shock" as proposed by Barbara McClintock [69] which usually results in a disruption of gene expression regulation and in the activation of genetic mobile elements, leading to new patterns of gene expression for part of genes in hybrids [70], as observed here in this study. We further show that the most prevalent expression profile when considering all sexes and crosses was the intermediate expression profile also referred to as additive [42,43]. These genes with an intermediate expression profile are associated with processes related to DNA integration and more specifically to retro-transposon activity (36% of retro-transposase were falling into intermediate expression profile), suggesting an expression of transposable elements in first generation hybrids that is different from the ones in the parental lines [71]. These transposable elements could be linked to the "genomic shock" expected in first generation hybrids [69]. However, this would need further investigation, starting with a complete annotation of these elements, to better decipher the control mechanisms that have been disrupted, as well as the impact that their expression might have in shaping the hybrid genomic landscape in subsequent generations.

The fact that among hybrid expression profiles the intermediate one was the most frequent also denotes that compatibility between S. haematobium and S. bovis is probably largely maintained. Intermediate expression of a gene in the hybrid offspring would be expected when the expression level of each parental allele is maintained (without interference due to different species origin) so their average expression at the gene level is indeed at an intermediate level. This would only be possible if the transcriptomic machineries, such as transcription factors and trans-regulatory factors, in S. haematobium and S. bovis, remain compatible one to the other. Schistosoma haematobium and S. bovis are generally known to present genomes that are highly permeable one to the other [24]. Recent studies tend to demonstrate various levels of introgression (up to 23% in the case of S. haematobium x S. bovis [23,25,41]) in many schistosome species [23,26,27,72,73]. At the transcriptomic level, mating between S. haematobium and S. bovis have also been found to trigger only few differential gene expressions which also suggests that the two species remain strongly co-adapted in terms of gene regulation [24]. In our case similarly, genes assigned to a particular expression profile were very specific to the hybrid sex and to the cross direction. Within a specific cross and sex, the actual number of genes that were not assigned to a profile actually represented the majority of genes (on average 81%). These unassigned genes could be either differentially expressed genes but not typical of any expression profiles or they could be genes that do not show any differential expression between either the two parental lines or the hybrids and parental lines. Thought, the investigation of such conserved genes in terms of expression was not the focus of the current study, we estimated that they corresponded to 20% to 26% (depending on the cross and the sex) of the total 15770 genes analysed. This showed that a substantial portion of the transcriptome was not

significantly impacted by hybridization. We thus suggest that the abundant number of genes unassigned to a particular hybrid expression profile along with the intermediate gene expression profiles being more prevalent among *S. haematobium* and *S. bovis* hybrids might be another facet of the signature of the high permeability that still exists between *S. haematobium* and *S. bovis* genomes.

Schistosoma first generation hybrids are not only fertile, but some offspring may show heterosis (higher infectivity, production of more numerous and larger eggs, [24,27,28,40]). Heterosis is commonly expected to result either from complementary expression of two inbred parental species (dominance model) or from synergetic interaction between parental alleles (over-dominance model) [9]. In terms of gene expression, each model is expected to translate either into over-expressed and under-expressed profiles or into parent-like expression profile respectively (i.e., overall non-intermediate expression profiles, [4,5,43,74,75] but see [42] for intermediate expression profiles associated with heterosis). In our study, we show that overand under-expressed profiles are more prevalent than parent-like expression profiles and thus contain more genes that are likely to be associated to heterosis in the case of Schistosoma hybrids. Consequently, if heterosis is confirmed in S. haematobium and S. bovis first generation hybrids, we suggest that at the molecular level it would be better explained by an overdominance model in which synergetic expression of parental alleles occurs. Furthermore, over- and under-expression profiles were more abundant in females than males (see Fig 2). In this study we were thus able to detect gene expression profiles compliant with heterosis expression in females, however, the lack of direct gene candidates for heterosis in male does not necessary mean that heterosis in S. haematobium x S. bovis male hybrids does not occur, including for instance at a different stage of parasite's the life cycle.

We further provide in this study some insights about the biological processes that are impacted in first generation hybrids compared to parental lines. Genes differentially expressed in hybrids were different depending on the sex and the cross, but they were overall involved in congruent biological processes across crosses and sexes within a particular hybrid expression profile. We already mentioned that genes having an intermediate expression profile were associated with DNA integration processes and potentially transposable elements activity which might relate to the genomic shock that hybrids undergo through an activation of mobile elements [69]. Some other biological processes from genes expressed differentially in hybrids could on the other hand be related to heterosis. Heterosis is indeed a complex multigenic trait for which no consensus has yet been found across the tree of life. Some authors suggest that it may be due to a cumulative effect of the differential expression of a variety of genes involved in one or several metabolic pathways affecting yield, energy use efficiency and/or reproductive success [76,77]. We found that Carbohydrate metabolic process was associated with genes expressed at an intermediate level in hybrids compared to parental lines. Carbohydrate metabolism could indeed relate to the parasite's ability to manage their food intake and/or regulation and use of energy. Similarly, we found that some biological processes that could be related to heterosis were enriched with the genes susceptible to lead to exacerbated expression of such processes in hybrids compared to parental lines (i.e., over-expressed genes but also underexpressed genes). First, among the experimental F1' females we found Reproductive process being over-expressed which could directly mirror observations made at the phenotypic level as in other Schistosoma species [78], although higher yields of eggs in S. haematobium x S. bovis first generation hybrids needs to be confirmed (but see [20]). Furthermore, non-coding RNA metabolic process was also enriched in both F1 and F1' females. Non-coding RNA have been suggested to greatly impact regulatory networks and were also directly associated with heterosis in rice hybrids [79]. Otherwise, regarding biological processes enriched with underexpressed genes, those were mostly related to signaling pathways (eg. Regulation of signaling

and *G-protein-coupled receptor signaling pathway, cell adhesion* and *Anatomical structure development*, Fig 4). While those processes are more difficult to interpret, one could speculate that these processes could relate to the mating process between the male and female. In particular, females are known to undergo strong developmental changes upon mating and this process relies on critical molecular cross talk between the two partners [80,81]. Under-expressing processes related to mating might contribute to ease future sex interactions within hybrids or with parental lines. The biological processes enriched in genes differentially expressed in first generation hybrids between *S. haematobium* and *S. bovis* can hence be linked to the genomic shock expected after hybridization but noticeably also to heterosis and compatibility of two co-adapted species.

It is worth noting that for a long time hybrid vigor has been studied independently from hybrid incompatibility. As such, studies aiming at identifying the molecular basis of heterosis are still often restricted to model species or species of agroeconomic interest [4-6,75]. However, some studies focusing on the molecular basis of hybrid incompatibilities have investigated wild species [82,83]. Interestingly both type of studies highlighted the presence of genes having nonadditive profiles but gave them a very contrasting interpretation (basis of heterosis on one hand and proof of mis-regulation in the other hand (eg., [6,83]). These observations along with our results seem to support the idea that heterosis is one facet of hybrid incompatibility. Here our findings can be interpreted at the light of the idea exposed in [7] stating that hybrid vigor can result from a mis-regulation of multiple process, potentially involved in trade-offs with reproduction and growth functions. Our results suggest a differential activation of transposable elements in hybrids compared to parental lines and those mobile elements are known to reshuffle gene regulatory networks [71]. On the other hand, we also identified congruent biological processes that can be directly associated with heterosis and that can reduce functional incompatibilities. We suspect DNA integration processes and transposable elements to be a consequence of hybridization that could contribute to establish new patterns of gene expression. The outcome of those new patterns of gene expression in hybrids can generally be on a continuum between heterosis and hybrid incompatibility but precisely understanding their adaptive effects in a particular environment should be the focus of further studies.

In conclusion, in this study we relied on a transcriptomic analysis of experimental first-generation hybrids between two *Schistosoma* species to unravel the molecular machinery of hybridization and the potential basis of the spectrum of hybrid incompatibility and heterosis. Our results are congruent with a maintained compatibility between the parental lines and also the probable occurrence of heterosis in first generation *S. haematobium* x *S. bovis* hybrids, potentially mediated through changes in gene regulation networks. We indeed found that 7,100 genes in hybrids were expressed differently than at least one of the parental species hence suggesting that changes in gene regulatory networks occur after hybridization. The predominance of hybrid genes expressed at intermediate levels compared to the parental lines highlights that compatibility between the parental species is still largely maintained. We also report that in *S. haematobium x S. bovis* first generation hybrids, gene expression patterns following an over-dominant model (hence compatible with heterosis through synergetic expression of each parental allele) included genes coding for functions related to reproductive success. Our study is hence meaningful in the context of recent discoveries around introgression events between *Schistosoma* species, while providing insights about what could occur in a scenario of ongoing hybridization.

Supporting information

S1 Fig. Principal Component Analysis based on 11 microsatellite markers (Sh9, Sh3, C102, Sh1, Sh14, C111, Sh13, Sh11, Sh2, Sh5 and Sh12). The blue dots are *S. haematobium* parasites

(miracidia) recovered from human in Cameroun [37]. The grey dots are adult *S. bovis* parasites collected from cattle [45]. The orange dots are the *S. haematobium* parasites (cercariae) that formed the strain at the origin of the crosses. First axis (68%)–Second axis (32%). (TIF)

S2 Fig. Principal Component Analysis of transcriptiomic data from *S. haematobium*, *S. bovis* or their first-generation reciprocal hybrids (F1: female *S. haematobium* x male *S. bovis*; F1: female *S. bovis* x male *S. haematobium*). We can see that samples cluster by sex (PC1) and parasite species origin (PC2). One male F1 sample clustered apart from others and has been excluded from further analysis.

(TIF)

S3 Fig. Heatmaps illustrating the gene clustering within each hybrid expression profiles in F1' females (F1') compared to *S. haematobium* (SH) and *S. bovis* (SB) females: under and over-expressed profiles (a), intermediate profiles (b), *S. haematobium*-like profile (c) and *S. bovis-like* profile (d).

(TIF)

S4 Fig. Heatmaps illustrating the gene clustering within each hybrid expression profiles in F1 males (F1) compared to *S. haematobium* (SH) and *S. bovis* (SB) males: under and over-expressed profiles (a), intermediate profiles (b), *S. haematobium*-like profile (c) and *S. bovis*-like profile (d).

(TIF)

S5 Fig. Heatmaps illustrating the gene clustering within each hybrid expression profiles in F1' males (F1') compared to *S. haematobium* (SH) and *S. bovis* (SB) males: under and over-expressed profiles (a), intermediate profiles (b), *S. haematobium*-like profile (c) and *S. bovis*-like profile (d).

(TIF)

S6 Fig. Venn plots showing shared genes involved in each hybrid expression profiles between the two hybrid crosses (F1 and F1') and the two sexes. (TIF)

S1 Table. Statistics extracted from gffcompare report when comparing our assembly to the reference genome of *S. haematobium*. (XLSX)

S2 Table. Details from the gffcompare output on the correspondence between transcripts and genes from our assembly and of the *S. haematobium*'s reference genome. At the transcript level are presented the number of transcripts that were found in the reference but not in our assembly (missing), the number of transcripts that exactly matched the full intron chain of transcripts of the refence (exact match), the number of transcripts that matched imperfectly (eg, not the full intron chain) transcripts of the reference (non-exact match), the number of transcripts that mapped unproperly to transcripts of the reference (unreliable match, i.e. including possible pre-mRNA fragment, match on the opposite strand likely resulting from a mapping error and no-actual overlap) and that did not match at all any transcripts of the reference (novel). The same statistics are provided at the gene level. We also present the total number of transcripts and gene in our assembly and in the reference genome of *S. haematobium*. (XLSX)

S1 File. Sheet A. Table summarising for each gene assembled from *S. haematobium*, *S. bovis* and their first-generation hybrids RNA-seq, their expression profile (if any), the cross and sex

in which their expression profile has been measured, relevant Log2 Fold Changes (LFC) for contrasts between hybrids and parents along with their associated False Discovery Rate (FDR). In bold are figured the LFC and FDR used when establishing a profile for a given gene. **Sheet B**. Table summarising for each gene assembled from *S. haematobium*, *S. bovis* and their first-generation hybrids RNA-seq, their expression profile (if any), the cross and sex in which their expression profile has been measured, GO terms annotated, putative gene product and (when any) the associated gene name and ID from the reference genome of *S. haematobium*. **Sheet C**. Table summarising the result of the gene ontology analysis and giving for each enriched Gene Ontology their associated expression profile, the level of the GO term, the total number of sequences having this GO term, the name of the GO term and their corrected p-value. (XLSX)

Acknowledgments

This study is set within the framework of the "Laboratoires d'Excellences (LABEX)": TULIP (ANR-10-LABX-41) and "CeMEB" (ANR-10-LABX-04-01).

Author Contributions

Conceptualization: Julien Kincaid-Smith, Jérôme Boissier, Eve Toulza.

Data curation: Eglantine Mathieu-Bégné.

Formal analysis: Eglantine Mathieu-Bégné.

Funding acquisition: Jérôme Boissier, Eve Toulza.

Investigation: Eglantine Mathieu-Bégné.

Methodology: Eglantine Mathieu-Bégné, Julien Kincaid-Smith, Cristian Chaparro, Eve Toulza.

Resources: Julien Kincaid-Smith, Jean-François Allienne.

Supervision: Jérôme Boissier, Eve Toulza.

Validation: Eglantine Mathieu-Bégné, Cristian Chaparro, Eve Toulza.

Visualization: Eglantine Mathieu-Bégné, Julien Kincaid-Smith.

Writing - original draft: Eglantine Mathieu-Bégné.

Writing - review & editing: Julien Kincaid-Smith, Olivier Rey, Jérôme Boissier, Eve Toulza.

References

- 1. Harrison RG, Larson EL. Hybridization, introgression, and the nature of species boundaries. Journal of Heredity. 2014; 15. https://doi.org/10.1093/jhered/esu033 PMID: 25149255
- 2. Baack EJ, Rieseberg LH. A genomic view of introgression and hybrid speciation. 2008; 9.
- Gompert Z, Buerkle CA. What, if anything, are hybrids: enduring truths and challenges associated with population structure and gene flow. 2016; 15. https://doi.org/10.1111/eva.12380 PMID: 27468308
- Xiao R, Yuan Y, Zhu F, He S, Ge Q, Wang X, et al. Transcriptomics and proteomics-based analysis of heterosis on main economic traits of silkworm, *Bombyx mori*. Journal of Proteomics. 2020; 229: 103941. https://doi.org/10.1016/j.jprot.2020.103941 PMID: 32805450
- Li S, Jayasinghege CPA, Guo J, Zhang E, Wang X, Xu Z. Comparative transcriptomic analysis of gene expression inheritance patterns associated with cabbage head heterosis. Plants. 2021; 10: 275. https:// doi.org/10.3390/plants10020275 PMID: 33572601

- Ponnanna K, DSouza SM, Ramachandra NB. *De novo* assembly, annotation and gene expression profiles of gonads of Cytorace-3, a hybrid lineage of *Drosophila nasuta nasuta* and *D. n. albomicans*. Genomics Inform. 2021; 19: e8. https://doi.org/10.5808/gi.20051 PMID: 33840172
- Bar-Zvi D, Lupo O, Levy AA, Barkai N. Hybrid vigor: The best of both parents, or a genomic clash? Current Opinion in Systems Biology. 2017; 6: 22–27. https://doi.org/10.1016/j.coisb.2017.08.004
- Birchler JA, Auger DL, Riddle NC. In search of the molecular basis of heterosis. The Plant Cell. 2003; 15: 2236–2239. https://doi.org/10.1105/tpc.151030 PMID: 14523245
- Lippman ZB, Zamir D. Heterosis: revisiting the magic. Trends in Genetics. 2007; 23: 60–66. <u>https://doi.org/10.1016/j.tig.2006.12.006</u> PMID: 17188398
- Šimková A, Civáňová K, Vetešník L. Heterosis versus breakdown in fish hybrids revealed by one-parental species-associated viral infection. Aquaculture. 2022; 546: 737406. <u>https://doi.org/10.1016/j.</u> aquaculture.2021.737406
- Mallet J. Hybridization as an invasion of the genome. Trends in Ecology & Evolution. 2005; 20: 229– 237. https://doi.org/10.1016/j.tree.2005.02.010 PMID: 16701374
- 12. King M. Species evolution: the role of chromosome change. Cambridge University Press; 1995.
- 13. Landry CR, Hartl DL, Ranz JM. Genome clashes in hybrids: insights from gene expression. Heredity. 2007; 99: 483–493. https://doi.org/10.1038/sj.hdy.6801045 PMID: 17687247
- King KC, Stelkens RB, Webster JP, Smith DF, Brockhurst MA. Hybridization in parasites: Consequences for adaptive evolution, pathogenesis, and public health in a changing world. Rall GF, editor. PLoS Pathog. 2015; 11: e1005098. https://doi.org/10.1371/journal.ppat.1005098 PMID: 26336070
- Greenspan SE, Lambertini C, Carvalho T, James TY, Toledo LF, Haddad CFB, et al. Hybrids of amphibian chytrid show high virulence in native hosts. Sci Rep. 2018; 8: 9600. <u>https://doi.org/10.1038/</u> s41598-018-27828-w PMID: 29941894
- Goodhead I, Capewell P, Bailey JW, Beament T, Chance M, Kay S, et al. Whole-genome sequencing of *Trypanosoma brucei* reveals introgression between subspecies that is associated with virulence. Carlton J, Fraser CM, editors. mBio. 2013; 4: e00197–13. https://doi.org/10.1128/mBio.00197-13 PMID: 23963174
- Fontaine MC, Pease JB, Steele A, Waterhouse RM, Neafsey DE, Sharakhov IV, et al. Extensive introgression in a malaria vector species complex revealed by phylogenomics. Science. 2015; 347: 1258524. https://doi.org/10.1126/science.1258524 PMID: 25431491
- Neafsey DE, Waterhouse RM, Abai MR, Aganezov SS, Alekseyev MA, Allen JE, et al. Highly evolvable malaria vectors: The genomes of 16 *Anopheles* mosquitoes. Science. 2015; 347: 1258522. <u>https://doi.org/10.1126/science.1258522</u> PMID: 25554792
- Llorens C, Arcos SC, Robertson L, Ramos R, Futami R, Soriano B, et al. Functional insights into the infective larval stage of *Anisakis simplex s.s.*, *Anisakis pegreffii* and their hybrids based on gene expression patterns. BMC Genomics. 2018; 19: 592. https://doi.org/10.1186/s12864-018-4970-9 PMID: 30086708
- Kincaid Smith J. Modification des traits d'histoire de vie au cours de l'hybridation et analyse des mécanismes moléculaires sous-jacents chez les parasites plathelminthes du genre Schistosoma. 2018.
- Fall CB, Lambert S, Léger E, Yasenev L, Garba AD, Diop SD, et al. Hybridized zoonotic Schistosoma infections result in hybridized morbidity profiles: a clinical morbidity study amongst co-infected human populations of Senegal. Microorganisms. 2021; 9: 1776. https://doi.org/10.3390/ microorganisms9081776 PMID: 34442855
- Colley DG, Bustinduy AL, Secor WE, King CH. Human schistosomiasis. Lancet (London, England). 2014; 383: 2253–2264. https://doi.org/10.1016/S0140-6736(13)61949-2 PMID: 24698483
- Platt RN, McDew-White M, Le Clec'h W, Chevalier FD, Allan F, Emery AM, et al. Ancient hybridization and adaptive introgression of an invadolysin gene in schistosome parasites. Shapiro B, editor. Molecular Biology and Evolution. 2019; 36: 2127–2142. https://doi.org/10.1093/molbev/msz154 PMID: 31251352
- 24. Kincaid-Smith J, Mathieu-Bégné E, Chaparro C, Reguera-Gomez M, Mulero S, Allienne J-F, et al. No pre-zygotic isolation mechanisms between *Schistosoma haematobium* and *Schistosoma bovis* parasites: From mating interactions to differential gene expression. Lamberton PHL, editor. PLoS Negl Trop Dis. 2021; 15: e0009363. https://doi.org/10.1371/journal.pntd.0009363 PMID: 33945524
- Rey O, Toulza E, Chaparro C, Allienne J-F, Kincaid-Smith J, Mathieu-Begné E, et al. Diverging patterns of introgression from *Schistosoma bovis* across *S. haematobium* African lineages. Steinauer M, editor. PLoS Pathog. 2021; 17: e1009313. https://doi.org/10.1371/journal.ppat.1009313 PMID: 33544762
- 26. Stothard JR, Kayuni SA, Al-Harbi MH, Musaya J, Webster BL. Future schistosome hybridizations: Will all *Schistosoma haematobium* hybrids please stand-up! Greenberg RM, editor. PLoS Negl Trop Dis. 2020; 14: e0008201. https://doi.org/10.1371/journal.pntd.0008201 PMID: 32614820

- Webster BL, Diaw OT, Seye MM, Webster JP, Rollinson D. Introgressive hybridization of *Schistosoma haematobium* group species in senegal: species barrier break down between ruminant and human schistosomes. PLoS Neglected Tropical Diseases. 2013; 7. https://doi.org/10.1371/journal.pntd. 0002110 PMID: 23593513
- Leger E, Webster JP. Hybridizations within the genus *Schistosoma*: implications for evolution, epidemiology and control. Parasitology. 2017; 144: 65–80. https://doi.org/10.1017/S0031182016001190 PMID: 27572906
- Southgate VR, Jourdane J, Tchuenté LAT. Invited review Recent studies on the reproductive biology of the schistosomes and their relevance to speciation in the Digenea. International Journal for Parasitology. 1998; 28: 1159–1172. https://doi.org/10.1016/S0020-7519(98)00021-6 PMID: 9762560
- Lockyer AE, Olson PD, Østergaard P, Rollinson D, Johnston DA, Attwood SW, et al. The phylogeny of the Schistosomatidae based on three genes with emphasis on the interrelationships of *Schistosoma* Weinland, 1858. Parasitology. 2003; 126: 203–224. https://doi.org/10.1017/S0031182002002792 PMID: 12666879
- Brémond P, Sellin B, Sellin E, Naméoua B, Labbo R, Théron A, et al. Arguments for the modification of the genome (introgression) of the human parasite *Schistosoma haematobium* by genes from *S. bovis*, in Niger. Comptes rendus de l'Academie des sciences Serie III, Sciences de la vie. 1993; 316: 667– 670.
- Huyse T, Webster BL, Geldof S, Stothard JR, Diaw OT, Polman K, et al. Bidirectional introgressive hybridization between a cattle and human schistosome species. PLoS Pathogens. 2009; 5. <u>https://doi.org/10.1371/journal.ppat.1000571</u> PMID: 19730700
- 33. Moné H, Holtfreter MC, Allienne JF, Mintsa-Nguéma R, Ibikounlé M, Boissier J, et al. Introgressive hybridizations of *Schistosoma haematobium* by *Schistosoma bovis* at the origin of the first case report of schistosomiasis in Corsica (France, Europe). Parasitology Research. 2015; 114: 4127–4133. https://doi.org/10.1007/s00436-015-4643-4 PMID: 26268566
- 34. Savassi BAES, Mouahid G, Lasica C, Mahaman S-DK, Garcia A, Courtin D, et al. Cattle as natural host for *Schistosoma haematobium* (Bilharz, 1852) Weinland, 1858 x *Schistosoma bovis* Sonsino, 1876 interactions, with new cercarial emergence and genetic patterns. Parasitol Res. 2020; 2189–2205. https://doi.org/10.1007/s00436-020-06709-0 PMID: 32468189
- Angora EK, Allienne J-F, Rey O, Menan H, Touré AO, Coulibaly JT, et al. High prevalence of Schistosoma haematobium × Schistosoma bovis hybrids in schoolchildren in Côte d'Ivoire. Parasitology. 2020; 147: 287–294. https://doi.org/10.1017/S0031182019001549 PMID: 31727202
- Onyekwere AM, Rey O, Allienne J-F, Nwanchor MC, Alo M, Uwa C, et al. Population genetic structure and hybridization of *Schistosoma haematobium* in Nigeria. Pathogens. 2022; 11: 425. <u>https://doi.org/ 10.3390/pathogens11040425</u> PMID: 35456103
- Teukeng FFD, Blin M, Bech N, Gomez MR, Zein-Eddine R, Simo AMK, et al. Hybridization increases genetic diversity in *Schistosoma haeamatobium* populations infecting humans in Cameroon. Infect Dis Poverty. 2022; 11: 37. https://doi.org/10.1186/s40249-022-00958-0 PMID: 35346375
- Boissier J, Grech-Angelini S, Webster BL, Allienne J-F, Huyse T, Mas-Coma S, et al. Outbreak of urogenital schistosomiasis in Corsica (France): an epidemiological case study. The Lancet Infectious Diseases. 2016; 16: 971–979. https://doi.org/10.1016/S1473-3099(16)00175-4 PMID: 27197551
- Wellinghausen N, Moné H, Mouahid G, Nebel A, Tappe D, Gabriel M. A family cluster of schistosomiasis acquired in Solenzara River, Corsica (France)—Solenzara River is clearly a transmission site for schistosomiasis in Corsica. Parasitol Res. 2022 [cited 5 Jul 2022]. https://doi.org/10.1007/s00436-022-07574-9 PMID: 35715618
- Huyse T, Van den Broeck F, Hellemans B, Volckaert FAM, Polman K. Hybridisation between the two major African schistosome species of humans. International Journal for Parasitology. 2013; 43: 687– 689. https://doi.org/10.1016/j.ijpara.2013.04.001 PMID: 23643461
- 41. Kincaid-Smith J, Tracey A, de Carvalho Augusto R, Bulla I, Holroyd N, Rognon A, et al. Morphological and genomic characterisation of the *Schistosoma* hybrid infecting humans in Europe reveals admixture between *Schistosoma haematobium* and *Schistosoma bovis*. Webster BL, editor. PLoS Negl Trop Dis. 2021; 15: e0010062. https://doi.org/10.1371/journal.pntd.0010062 PMID: 34941866
- **42.** Guo M, Rupe MA, Yang X, Crasta O, Zinselmeier C, Smith OS, et al. Genome-wide transcript analysis of maize hybrids: allelic additive gene expression and yield heterosis. Theor Appl Genet. 2006; 113: 831–845. https://doi.org/10.1007/s00122-006-0335-x PMID: 16868764
- Wu X, Li R, Li Q, Bao H, Wu C. Comparative transcriptome analysis among parental inbred and crosses reveals the role of dominance gene expression in heterosis in *Drosophila melanogaster*. Sci Rep. 2016; 6: 21124. https://doi.org/10.1038/srep21124 PMID: 26928435

- Tchuem Tchuenté L-A, Rollinson D, Stothard JR, Molyneux D. Moving from control to elimination of schistosomiasis in sub-Saharan Africa: time to change and adapt strategies. Infect Dis Poverty. 2017; 6: 42. https://doi.org/10.1186/s40249-017-0256-8 PMID: 28219412
- 45. Djuikwo-Teukeng FF, Kouam Simo A, Allienne J-F, Rey O, Njayou Ngapagna A, Tchuem-Tchuente LA, et al. Population genetic structure of *Schistosoma bovis* in Cameroon. Parasites Vectors. 2019; 12: 56. https://doi.org/10.1186/s13071-019-3307-0 PMID: 30678712
- Silva M, Vicente F, Avelino I, Martin V. Susceptibility of *Planorbarius metidjensis* from Portugal and Spain to *Schistosoma bovis* from Salamanca, Spain. Malacologia. 1977; 16: 251–254.
- 47. Kincaid-Smith J, Boissier J, Allienne J-F, Oleaga A, Djuikwo-Teukeng F, Toulza E. A Genome wide comparison to identify markers to differentiate the sex of larval stages of *Schistosoma haematobium*, *Schistosoma bovis* and their respective hybrids. Gasser RB, editor. PLoS Negl Trop Dis. 2016; 10: e0005138. https://doi.org/10.1371/journal.pntd.0005138 PMID: 27861520
- Théron A, Pages J-R, Rognon A. Schistosoma mansoni: Distribution patterns of miracidia among Biomphalaria glabrata snail as related to host susceptibility and sporocyst regulatory processes. Experimental Parasitology. 1997; 85: 1–9. https://doi.org/10.1006/expr.1996.4106 PMID: 9024196
- Boissier J, Rivera ER, Moné H. Altered behavior of the snail *Biomphalaria glabrata* as a result of infection with *Schistosoma mansoni*. Journal of Parasitology. 2003; 89: 429–433. https://doi.org/10.1645/0022-3395(2003)089[0429:ABOTSB]2.0.CO;2
- Boissier J, Chlichlia K, Digon Y, Ruppel A, Moné H. Preliminary study on sex-related inflammatory reactions in mice infected with *Schistosoma mansoni*. Parasitology Research. 2003; 91: 144–150. https:// doi.org/10.1007/s00436-003-0943-1 PMID: 12910415
- Stroehlein AJ, Korhonen PK, Lee VV, Ralph SA, Mentink-Kane M, You H, et al. Chromosome-level genome of *Schistosoma haematobium* underpins genome-wide explorations of molecular variation. Zamanian M, editor. PLoS Pathog. 2022; 18: e1010288. <u>https://doi.org/10.1371/journal.ppat.1010288</u> PMID: 35167626
- Young ND, Jex AR, Li B, Liu S, Yang L, Xiong Z, et al. Whole-genome sequence of Schistosoma haematobium. Nat Genet. 2012; 44: 221–225. https://doi.org/10.1038/ng.1065 PMID: 22246508
- Oey H, Zakrzewski M, Gravermann K, Young ND, Korhonen PK, Gobert GN, et al. Whole-genome sequence of the bovine blood fluke *Schistosoma bovis* supports interspecific hybridization with *S. haematobium*. Platt RN, editor. PLoS Pathog. 2019; 15: e1007513. <u>https://doi.org/10.1371/journal.ppat</u>. 1007513 PMID: 30673782
- Bolger AM, Lohse M, Usadel B. Trimmomatic: a flexible trimmer for Illumina sequence data. Bioinformatics. 2014; 30: 2114–2120. https://doi.org/10.1093/bioinformatics/btu170 PMID: 24695404
- Kim D, Langmead B, Salzberg SL. HISAT: a fast spliced aligner with low memory requirements. Nat Methods. 2015; 12: 357–360. https://doi.org/10.1038/nmeth.3317 PMID: 25751142
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The Sequence Alignment/Map format and SAMtools. Bioinformatics. 2009; 25: 2078–2079. <u>https://doi.org/10.1093/bioinformatics/btp352</u> PMID: 19505943
- Pertea M, Pertea GM, Antonescu CM, Chang T-C, Mendell JT, Salzberg SL. StringTie enables improved reconstruction of a transcriptome from RNA-seq reads. Nat Biotechnol. 2015; 33: 290–295. https://doi.org/10.1038/nbt.3122 PMID: 25690850
- 58. Pertea G, Pertea M. GFF Utilities: GffRead and GffCompare [version 1; peer review: 2. 2020.
- Liao Y, Smyth GK, Shi W. featureCounts: an efficient general purpose program for assigning sequence reads to genomic features. Bioinformatics. 2014; 30: 923–930. <u>https://doi.org/10.1093/bioinformatics/ btt656 PMID: 24227677</u>
- R Core Team. R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria; 2020. Available: URL https://www.R-project.org/.
- Love MI, Huber W, Anders S. Moderated estimation of fold change and dispersion for RNA-seq data with DESeq2. Genome Biol. 2014; 15: 550. <u>https://doi.org/10.1186/s13059-014-0550-8</u> PMID: 25516281
- Giardine B, Riemer C, Hardison RC, Burhans R, Elnitski L, Shah P, et al. Galaxy: A platform for interactive large-scale genome analysis. Genome Res. 2005; 15: 1451–1455. <u>https://doi.org/10.1101/gr.</u> 4086505 PMID: 16169926
- Goecks J, Nekrutenko A, Taylor J, Galaxy Team T. Galaxy: a comprehensive approach for supporting accessible, reproducible, and transparent computational research in the life sciences. Genome Biol. 2010; 11: R86. https://doi.org/10.1186/gb-2010-11-8-r86 PMID: 20738864
- 64. Haas B, Papanicolaou A, Yassour M. TransDecoder. TransDecoder. 2017.

- Jones P, Binns D, Chang H-Y, Fraser M, Li W, McAnulla C, et al. InterProScan 5: genome-scale protein function classification. Bioinformatics. 2014; 30: 1236–1240. <u>https://doi.org/10.1093/bioinformatics/</u> btu031 PMID: 24451626
- Dainat J. Another Gtf/Gff Analysis Toolkit (AGAT): Resolve Interoperability Issues and Accomplish More with Your Annotations. PAG; 2022.
- Wright RM, Aglyamova GV, Meyer E, Matz MV. Gene expression associated with white syndromes in a reef building coral, *Acropora hyacinthus*. BMC Genomics. 2015; 16: 371–383. https://doi.org/10.1186/ s12864-015-1540-2 PMID: 25956907
- Supek F, Bošnjak M, Škunca N, Šmuc T. REVIGO summarizes and visualizes long lists of gene ontology terms. Gibas C, editor. PLoS ONE. 2011; 6: e21800. https://doi.org/10.1371/journal.pone.0021800 PMID: 21789182
- 69. McClintock B. The significance of responses of the genome to challenge. Science. 1984; 226: 792–801. https://doi.org/10.1126/science.15739260 PMID: 15739260
- 70. Ha M, Lu J, Tian L, Ramachandran V, Kasschau KD, Chapman EJ, et al. Small RNAs serve as a genetic buffer against genomic shock in *Arabidopsis* interspecific hybrids and allopolyploids. Proc Natl Acad Sci USA. 2009; 106: 17835–17840. https://doi.org/10.1073/pnas.0907003106 PMID: 19805056
- Rebollo R, Romanish MT, Mager DL. Transposable elements: an abundant and natural source of regulatory sequences for host genes. Annu Rev Genet. 2012; 46: 21–42. <u>https://doi.org/10.1146/annurev-genet-110711-155621 PMID: 22905872</u>
- 72. Savassi BAES, Dobigny G, Etougbétché JR, Avocegan TT, Quinsou FT, Gauthier P, et al. *Mastomys natalensis* (Smith, 1834) as a natural host for *Schistosoma haematobium* (Bilharz, 1852) Weinland, 1858 x *Schistosoma bovis* Sonsino, 1876 introgressive hybrids. Parasitol Res. 2021; 120: 1755–1770. https://doi.org/10.1007/s00436-021-07099-7 PMID: 33687566
- Berger DJ, Léger E, Sankaranarayanan G, Sène M, Diouf ND, Rabone M, et al. Genomic evidence of contemporary hybridization between *Schistosoma* species. Zamanian M, editor. PLoS Pathog. 2022; 18: e1010706. https://doi.org/10.1371/journal.ppat.1010706 PMID: 35939508
- 74. Li A, Fang M-D, Song W-Q, Chen C-B, Qi L-W, Wang C-G. Gene expression profiles of two intraspecific Larix lines and their reciprocal hybrids. Mol Biol Rep. 2012; 39: 3773–3784. https://doi.org/10.1007/ s11033-011-1154-y PMID: 21750915
- 75. Ng WL, Wu W, Zou P, Zhou R. Comparative transcriptomics sheds light on differential adaptation and species diversification between two *Melastoma* species and their F1 hybrid. AoB PLANTS. 2019; 11. https://doi.org/10.1093/aobpla/plz019 PMID: 31037213
- 76. Baranwal VK, Mikkilineni V, Zehr UB, Tyagi AK, Kapoor S. Heterosis: emerging ideas about hybrid vigour. Journal of experimental botany. 2012; 63: 6309–6314. <u>https://doi.org/10.1093/jxb/ers291</u> PMID: 23095992
- 77. Goff SA. A unifying theory for general multigenic heterosis: Energy efficiency, protein metabolism, and implications for molecular breeding. New Phytologist. 2011. pp. 923–937. <u>https://doi.org/10.1111/j.</u> 1469-8137.2010.03574.x PMID: 21166808
- 78. Wright C, Ross G. Hybrids between Schistosoma haematobium and S. mattheei and their identification by isoelectric focusing of enzymes. Transactions of the Royal Society of Tropical Medicine and Hygiene. 1980; 74: 326–332.
- 79. Wang M, Wang J. Non-coding RNA expression analysis revealed the molecular mechanism of flag leaf heterosis in inter-subspecific hybrid rice. Front Plant Sci. 2022; 13: 990656. <u>https://doi.org/10.3389/fpls.</u> 2022.990656 PMID: 36226282
- Lu Z, Sessler F, Holroyd N, Hahnel S, Quack T, Berriman M, et al. Schistosome sex matters: a deep view into gonad-specific and pairing-dependent transcriptomes reveals a complex gender interplay. Sci Rep. 2016; 6: 31150. https://doi.org/10.1038/srep31150 PMID: 27499125
- Grevelding CG, Sommer G, Kunz W. Female-specific gene expression in *Schistosoma mansoni* is regulated by pairing. Parasitology. 1997; 115: 635–640. <u>https://doi.org/10.1017/S0031182097001728</u> PMID: 9488875
- 82. McGirr JA, Martin CH. Ecological divergence in sympatry causes gene misexpression in hybrids. Mol Ecol. 2020; 29: 2707–2721. https://doi.org/10.1111/mec.15512 PMID: 32557903
- 83. Sánchez-Ramírez S, Weiss JG, Thomas CG, Cutter AD. Widespread misregulation of inter-species hybrid transcriptomes due to sex-specific and sex-chromosome regulatory evolution. Buerkle A, editor. PLoS Genet. 2021; 17: e1009409. https://doi.org/10.1371/journal.pgen.1009409 PMID: 33667233